Author’s response to reviews

Title: Do health care professionals' perceptions help to measure the degree of overcrowding in the emergency department? A pilot study in an Italian University Hospital

Authors:

Giorgia Valpiani (giorgia.valpiani@ospfe.it)
Strada Andrea (a.strada@ospfe.it)
Francesca Bravi (f.bravi@ospfe.it)
Roberto Bentivegna (r.bentivegna@ospfe.it)
Tiziano Carradori (t.carradori@ospfe.it)

Version: 2 Date: 01 Jul 2019

Author’s response to reviews:

Ferrara 28/06/2019

Subject: Submission of revised paper EMMD-D-19-00015R1 (Manuscript Number)

Dear Editor,

Please find enclosed our manuscript, “Do health care professionals' perceptions help to measure the degree of overcrowding in Emergency Department? A pilot study in an Italian University Hospital” by Andrea Strada et al., which we would like to submit for publication as research article in BMC Emergency Medicine.

Thank you for your suggestions, we have integrated all comments in the manuscript. We are returning the above paper, revised on the basis of the reviewers’ remarks. We would like to thank the reviewer for careful and thorough reading of this manuscript and for the thoughtful comments and constructive suggestions, which help to improve the quality of this manuscript.
Looking forward to hearing from you, we remain,

Yours sincerely,

Giorgia Valpiani

Please address all correspondence to:

Giorgia Valpiani
University Hospital Trust Ferrara
Via Aldo Moro 8 44124 Ferrara (Italia)
Telephone : 0039 0532 237 820
giorgia.valpiani@ospfe.it
ORCID ID 0000-0002-1401-8012